Author's response to reviews

Title: The International Documentation and Evaluation System IDES: a single center observational case series for development of an ankle prosthesis documentation questionnaire and study of its feasibility and face validity

Authors:

Peter Diel (peter_diel@insel.ch)
Christoph Thier (christophthier@web.de)
Emin Aghayev (emin.aghayev@memcenter.unibe.ch)
Markus Preis (preis@orthopaedie-aukamm.de)
Marcel Dudda (marcel.dudda@ruhr-uni-bochum.de)
Norman Espinosa (norman.espinosa@balgrist.ch)
Christoph Röder (christoph.roeder@memcenter.unibe.ch)

Version: 3 Date: 8 March 2010

Author's response to reviews:

Author list: the list of authors in the manuscript adapted to your preferred style ?First name, Initial, Last name? (e.g. Joe F Bloggs)

Affiliations: Each affiliation was written in full according to your proposed format: Department, Institute, (Street), City, (Postal code), Country.

Authors' contributions: We have now indicated that all authors read and approved the final manuscript.

Figures: All figures were adapted and renumbered. The figures which had two pages are now converted to one figure-file including both pages.

Citation for all figures, at least once, is provided.

The word count and keywords were deleted from the abstract and the word count was also deleted from the discussion.

The competing interests and authors contributions sections were moved at the end of the manuscript, just before the references.

Figure legends were moved after the references.

All figure citations were checked and adapted to the full word "figure".

References were cited in the text before the full stop. For example ?text [1].?

Typography was checked.

Kind regards,

E. Aghayev